Author’s response to reviews

Title: Compliance with herpes zoster vaccination in young and adult individuals in two regions of Italy

Authors:

Antonino Parlato (antoninoparlato1@virgilio.it)
Vincenzo Romano Spica (vincenzo.romanospica@iusm.it)
Massimo Ciccozzi (ciccozzi@iss.it)
Francesca Farchi (francesca.farchi@iss.it)
Francesca Gallè (francesca.galle@uniparthenope.it)
Valeria Di Onofrio (valeriadionofrio@libero.it)
Elisabetta Franco (franco@med.uniroma2.it)
Giorgio Liguori (giorgio.liguori@uniparthenope.it)

Version: 6 Date: 8 June 2010

Author’s response to reviews:

Dear Sirs,

we have modify the text on the basis of your requests.

We hope the manuscript could be now ready for publication.

Best regards

Giorgio Liguori